Author’s response to reviews

Title: Outcome prediction in disorders of consciousness: the role of coma recovery scale revised

Authors:
Lucia Lucca (l.lucca@istitutosantanna.it)
Danilo Lofaro (danilo.lofaro@unical.it)
Loris Pignolo (l.pignolo@isakr.it)
Elio Leto (e.leto@istitutosantanna.it)
Maria Ursino (m.ursino@istitutosantanna.it)
Maria Daniela Cortese (d.cortese@istitutosantanna.it)
Domenico Conforti (domenico.conforti@unical.it)
Paolo Tonin (patonin18@gmail.com)
antonio cerasa (a.cerasa@unicz.it)

Version: 5 Date: 28 Mar 2019

Author’s response to reviews:

Editor Comments:

1. While assessing your manuscript we found several instances where the text displayed overlap with other previously published works, in particular:
doi: 10.1586/ern.10.148
Overlap with this previously published work was found mainly in lines 11 – 13 of page 2.

And

https://www.nature.com/articles/nrneurol.2013.279?foxtrotcallback=tru
Overlap with this previously published work was found mainly in lines 7 – 10 of page 7.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. We would therefore be grateful if you could reformulate the aforementioned section of your paper to resolve the overlap between your manuscript and other sources. Please ensure that, where relevant,
these sources are also referenced as appropriate.

If this study uses methodology from a previously published work, please provide a summarizing statement in the methodology together with a citation to the original paper.
REPLY: These two instances have been completely re-formulated

2. Rather than uploading your title page separately, please include it at the beginning of your main manuscript file.
REPLY: DONE

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
REPLY: DONE